

# Temi di discussione

(Working papers)

What determines the size of bank loans in industrialized countries? The role of government debt

by Riccardo De Bonis and Massimiliano Stacchini



The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Patrizio Pagano, Alfonso Rosolia, Ugo Albertazzi, Claudia Biancotti, Giulio Nicoletti, Paolo Pinotti, Enrico Sette, Marco Taboga, Pietro Tommasino, Fabrizio Venditti.

Editorial Assistants: ROBERTO MARANO, NICOLETTA OLIVANTI.

#### WHAT DETERMINES THE SIZE OF BANK LOANS IN INDUSTRIALIZED COUNTRIES? THE ROLE OF GOVERNMENT DEBT

by R. De Bonis<sup>\*</sup> and M. Stacchini<sup>\*</sup>

#### Abstract

Given the importance of banking intermediation, we investigate the determinants of the size of bank loans in 18 OECD countries in the period 1981-1997. The aim of the paper is to show that the ratio of government debt to GDP has a negative effect on the level of bank credit. Second, countries with a German legal origin have higher ratios of loans to GDP than common law countries. Our results are robust to including such variables in the regressions as per capita GDP, stock market capitalization, the banking reserve requirement, the level of inflation and its volatility, openness to trade and the use of different econometric methods.

JEL classification: G21; G18; C23.

Keywords: bank loans, government debt, financial repression, legal origin of finance.

#### Contents

1. Introduction	5
2. A review of the literature	6
3. The data and some stylized facts	8
4. The empirical specification and the econometric results	10
4.1 The empirical model	10
4.2 Determinants of the ratio of bank loans to GDP	11
5. Robustness checks	12
5.1 Do annual data regressions replicate the five-year average statistics results?	12
5.2 Splitting the period into two sub-periods	13
5.3 Creditor rights vs. legal origin	13
5.4 Do results survive if corporate bonds are included in the picture ?	15
5.5 Other robustness checks	15
6. Conclusions	17
Tables and figures	
References	

<sup>\*</sup> Bank of Italy, Economics, Research and International Relations.

#### **1. Introduction**<sup>1</sup>

Banks are a key element of financial structures all over the world, contributing to the financing of the economy and the management of the payment system. But why are banks more important in some countries and less so in others? In 2004 the ratio of loans to GDP was around 46 per cent in the US, 77 per cent in France and 100 per cent in Germany. The ratio of deposits to GDP was 40 per cent in the US, 68 per cent in France and 86 per cent in Spain.

Different strands of literature studied how the level of bank loans is influenced by economic and institutional variables. Since the 1960s the financial repression approach rationalized the restrictions that affected the financial systems of industrialized countries (McKinnon, 1973; Shaw, 1973). These studies refer to policy instruments – such as high reserve requirement, implicit taxes levied through inflation, explicit taxation, issues of government bonds – that might reduce the size of banking systems. On the contrary, path dependency theories of financial development stress the role of legal origin as the central factor driving the cross-country differences we observe today (La Porta et al, 1997). The idea that legal origin is the major force explaining financial structures has been criticized as being excessively mechanistic. Financial systems change because of the evolution of the real economies and new political choices. For example Rajan and Zingales (2003) showed how financial markets in civil law countries were reversed throughout the last century and start to recover only at the end of the 1980s. Other authors state antagonism between banks and stock markets. This is a very old argument whereas other studies postulate a positive nexus between banks and equity markets.

The goal of this paper is to present new evidence on the determinants of bank loans in OECD countries, discriminating between the different factors by placing them in a general empirical model. The issue we address has not only a theoretical but also a policy relevance. If factors that limit bank development are detected, political authorities may introduce reforms to foster bank credit, whose link with economic growth has been established by many

<sup>&</sup>lt;sup>1</sup> We would like to thank Giovanni Ferri, Silvia Magri, Franco Peracchi, Enrico Perotti, Lisa Rodano, Federico Signorini, Damiano Silipo, Robert Waldmann and two anonymous referees for discussions and helpful comments on previous versions of this paper. We also thank the participants at seminars held at the Universities of Ancona, Bari, Cassino and Ferrara, at the XIV International "Tor Vergata" Conference on Banking and Finance, Rome, 5-7 December 2005 and at the 39<sup>th</sup> Annual Conference of the Money Macro and Finance Research Group, 12-14 September 2007, held at the University of Birmingham. The views are those of the authors and do not necessarily reflect those of the Bank of Italy.

contributions. As far as we know, our new insight is the negative effect of government debt on bank loans.

While most of the literature concentrates on cross-sections, we present a panel analysis of the main OECD countries in the period 1981-1997. Cross-sections for large samples of heterogeneous countries run the risk of not taking into account unobservable factors that may influence the results. Developed countries are more homogeneous and therefore easier to compare.

The paper is divided into 6 sections: the following Section 2 reviews the literature; Section 3 presents the data and introduces some stylized facts; Section 4 illustrates the empirical specification and comments on the econometric exercises; Section 5 contains some robustness checks; and Section 6 concludes.

#### 2. A review of the literature

Our paper is connected to various strands of the literature. A first link is with the theory of financial repression. This approach emphasizes a supply side aspect of financial development: governments may repress the size of the financial and banking sectors to raise revenues from alternative channels that lie under their control. Financial repression artificially increases the private sector's demand for those assets that are the basis of implicit taxes, e.g. the monetary base and the public debt. By the second half of the 1960s, regulations, taxes, qualitative and quantitative controls, often introduced to pursue monetary policy goals, made it difficult for financial intermediaries to fulfil their technological potential (Roubini and Sala-i-Martin, 1995). These measures shrank the financial systems of many countries (Battilossi, 2003). Here we review some of the contributions on these subjects.

Following Fisher's ideas on debt deflation, Boyd et al (2001) assess that inflation causes lower rates of return on loans, thus leading to reduced credit. Smaller real returns can not only decrease the availability of financing but also draw lower quality borrowers into the pool of credit seekers. The authors find a negative relationship between inflation and banking sector size.

There are links between inflation, reserve requirement and government bonds. Brock (1989) considers not only the inflation tax levied on currency but also that on banks' noninterest bearing reserves. In addition, regulation of banks' portfolio composition is a complementary instrument of financial repression. The government may impose portfolio constraints by setting public debt securities as the only assets eligible for meeting banks' requirements: as a consequence banks' resources are channeled towards the purchase of government bonds. Further, the government may impose credit ceilings as demonstrated by many industrialized countries in the 1970s. Giovannini and De Melo (1991) explain how nominal interest rate ceilings combined with controls on international flows may generate an artificially low cost of domestic funding for the government and consequent revenues from financial repression. However the link between fiscal policy and banking is not entirely clear. Studying the Italian case, Piluso and Ricciuti (2008) find evidence of pro-cyclical behaviour as regards public spending, taxes and bank variables, even if the causal connections are not investigated.

Our paper also looks at the law and finance view. This approach emphasizes the idea that early institutions and arrangements persist over time and determine current outcomes. The mechanism rests on two points: on the one hand, the basis of finance is the protection of investor and creditor rights; on the other hand, the legal origin of countries is the source of this basis. Legal systems are classified into four categories of origin: Anglo-Saxon, French, German and Scandinavian. They emerged in previous centuries, spreading abroad through conquest and imitation and can explain the differences in investor protection, contracting environment and financial development that countries exhibit today. La Porta et al (1997) show that countries with higher shareholders rights and common law traditions, by and large nations of Anglo-Saxon legal origin, have relatively larger financial markets. The legal approach has been criticized because it overlooks structural breaks in history. Beck et al (2003a) compared the legal traditions paradigm with the "adaptability channel of law": the authors find that legal origin matters for financial development because legal traditions differ in their ability to adapt efficiently to evolving economic conditions. They conclude that in terms of adaptability the German system is closer to the common law than the French legal system. We will come back to this issue of the nature of the German legal system later on.

Other scholars criticized the legal origin approach, underlining the role of interest groups changing over time in explaining cross-country differences in financial systems. Rajan and Zingales (2003) say that preferences of influential interest groups were implemented through the last century by the choice of opening to trade: they show how both in 1913 and in 1997, for any level given of demand, financial development was higher in countries that were more open to trade.<sup>2</sup> The idea is that international competition weakens the incumbents' opposition to financial development. Quy-Toan Do and Levchenko (2004) provide empirical evidence that trade openness affects the financial systems of countries in a non linear way:

 $<sup>^{2}</sup>$  Alcalà and Ciccone (2003) show how trade openness was a robust determinant of economic growth in the period 1960-1996.

among developed countries trade promotes financial systems while it has an opposite effect among the poorer ones. Chinn and Ito (2006) focus on the effects of capital account liberalization on financial growth and conclude that financial openness matters only when a threshold level of legal and institutional development has been attained.

In Section 3 we present our empirical strategy.

#### 3. The data and some stylized facts

We examine the effects of forms of financial repression and legal origin on bank loans, controlling for stock market capitalization, per capita income and other indicators. We use both contemporaneous variables (such as the reserve requirement and public debt), lagged variables (such as per capita GDP) and time-invariant variables (such as legal origin) referred to 18 OECD countries. The analysis is based upon data originally recorded at an annual frequency, over the period 1981-1997. We chose this time interval because the first composition of the euro area was defined in 1998 and this institutional break might have affected both the behaviour of private credit and its statistical definition.<sup>3</sup>

With regard to our dependent variable, we use the definition adopted by Levine and Zervos (1998) and Levine et al (2000), i.e. the credit granted by the banking system to the private sector, relative to GDP.

Moving to the independent variables, financial repression is measured through different indicators. The reserve requirement is proxied by the ratio of a bank's liquid reserves to its assets. Inflation rates are measured as annual growth rates of the consumer price index. Inflation volatility, measured by the annual standard deviation of the monthly inflation rate, might also affect credit. Public debt is measured by the gross general government debt as a percentage of GDP. It is difficult to collect statistics on other financial repression instruments, such as quantitative ceilings or controls on deposit and lending rates. Moreover most of these measures were gradually suppressed in OECD countries in the 1980s.

Legal origin is considered using the dummies introduced by La Porta et al (1997). The same source was exploited to assess the level of creditor rights protection. Trade openness is measured with the index suggested by Rajan and Zingales (2003), i.e. the sum of exports and imports of goods and services as a share of GDP.

<sup>&</sup>lt;sup>3</sup> The creation of the euro area implied a new definition of many items of bank balance sheets in member states. Moreover in recent years securitization has made the comparison of bank credit in different countries more difficult.

According to a large, though still controversial, body of literature, there is a link between economic growth and finance. While the direction of causality is difficult to establish, the predominant empirical finding is that financial development causes economic growth (for a survey, see Levine, 2003). There is also an opposing view that was probably initiated by Joan Robinson's claim that "finance follows". In this interpretation the structure of the economy determines the types of intermediaries that are best suited to industrial firms. Allen et al (2006) show that financial institutions and markets develop in response to the needs of firms and the characteristics of the real economy. We take this debate into account by using per capita GDP as a proxy of the demand side schedule which might influence the equilibrium level of bank credit. Per capita income is measured in terms of purchasing power parity.

Finally, other studies look at financial systems emphasizing the differences between bank-based and market-based countries. This approach implies that banks and markets are substitutes: households may prefer to invest in deposits rather than in shares; firms may favour shares over bank loans. In a different perspective, stock market capitalization may promote banking activity if banks and markets play a complementary role (Levine and Zervos, 1998). In this view both intermediaries and markets matter for growth (see Demirgüç -Kunt and Levine, 2001, for a synthesis). Given the existence of different opinions, we control how stock market capitalization, as a share of GDP, is linked to banking intermediation.

Table 1 reports the data definition and the sources of the statistics, while Table 2 lists the countries we selected, mainly on the basis of the available long term time series. Table 3 presents some descriptive statistics. Over all periods and all countries, the average ratios of loans to GDP is 0.71. As indicated by minimum and maximum values, there are marked differences across countries and years for both bank loans and explanatory variables.





9

Figure 1 shows the link between countries' private credit and legal traditions. Banking development is substantially greater in the German origin group than in countries with an Anglo-Saxon legal tradition. Further, all Scandinavian countries are located below the average size. This evidence is confirmed by the correlation matrix (Table 4), which shows a positive association between banking development and countries belonging to the German legal tradition while a negative correlation is detected for Scandinavian countries. The matrix also shows a negative correlation between government debt and the size of bank loans. Negative linkages are observed for bank credit and either inflation rates or reserve requirement. Further, there is a positive association between bank credit, on the one hand, and both stock market capitalization and per capita income, on the other.

In the following section we turn to multivariate regressions.

#### 4. The empirical specification and the econometric results

#### 4.1 The empirical model

The panel nature of the dataset is exploited through a static model specified as:

$$priv_{it} = \alpha + \beta X_{it} + \eta_i + \varepsilon_{it}$$

where  $priv_{it}$  are loans to the private sector, the vector  $X_{it}$  contains time-varying determinants (such as financial repression instruments, per capita GDP and stock market capitalization) while the term  $\eta_i$  captures "path dependency" time-invariant country components.

We followed the standard practice of taking five-year averages to remove the effects of the business cycle from the analysis (see Islam, 1995). As a further control, we estimated our models over the original annual database (see paragraph 5.1). To avoid the endogeneity issues associated with reverse causality between banking activity, on the one hand, and per capita income and stock market capitalization, on the other, we specified those explanatory variables also through lagged terms.<sup>4</sup> In such a way they are predetermined with respect to the dependent variable.

<sup>&</sup>lt;sup>4</sup> When we investigate five-year average statistics, those explanatory variables are instrumented through fiveyears average lagged terms. On the contrary, when annual data are analysed, the variables are instrumented through one-year lagged terms.

Cross-country heterogeneity - not captured by financial repression time-varying components - was taken into account through legal origin dummy variables, the first difference estimator, or by estimating both random and fixed effect models<sup>5</sup>.

#### 4.2 Determinants of the ratio of bank loans to GDP

In studying the determinants of the credit granted by the banking system to the private sector, as a share of GDP, we follow a step-by-step procedure, starting from a simple model and then adding new variables. Overall we present seven different regressions (Table 5). Beginning with the financial repression variables, the inflation rate sometimes has a negative impact on banking loans, but the coefficients are never significant. Moreover the volatility of inflation does not affect the ratio of credit to GDP. The coefficient of the reserve requirement has a negative sign but it is significant only in two regressions. We cannot support the results of Haslagh and Koo (1999) that the reserve requirement has a negative impact on banking business.

On the contrary, government debt has a negative and statistically significant influence on the ratio of loans to GDP. A 10 per cent increase in the government debt to GDP ratio is expected to lower credit granted to private sectors by 3 (regression 2) or 5 (regression 7) percentage points, respectively. Our result might reflect two similar and interdependent phenomena. According to the public finance approach to financial repression (Giovannini and De Melo, 1991; Battilossi, 2003; Roubini and Sala-i-Martin, 1995), issuing government securities is a way for the State to collect revenues, especially when the proceeds from legal taxation are difficult or costly to obtain, and banks find government securities a good investment. The second interpretation is a typical crowding out effect: in countries where the government has a large involvement in the economy, greater shares of bank credit may flow to State-owned firms, reducing the loans to the private sector. Our results are consistent with those obtained by Hauner (2008), who finds a negative link between public debt held by banks and financial development in middle-income countries.

While the financial repression literature brings into prominence the distortions that affected the banking systems since the 1960s, the legal origin theory emphasizes an older historical derivation of financial systems. Our exercises show that a share of cross-country variability is captured by the legal origin components. The significance of the legal origin variables is seen both in regression 6 (where individual random components are allowed) and

<sup>&</sup>lt;sup>5</sup> In general terms, when cross-country heterogeneity - not accounted for by time-varying variables – is captured by legal origin of countries, pooled estimators with legal dummies are more efficient than fixed effect models.

in regression 7 (where legal origin captures all the variability not accounted for by timevarying determinants). Countries based on German or French law have greater ratios of loans to GDP than common law countries. The German origin effect is the strongest. German origin countries have around 50 percentage points higher ratios of private credit to GDP with respect to the common law countries, while French origin countries have around 30 percentage points greater ratios of loans to GDP with respect to the common law countries. The role of legal origin in influencing capital market development has been recognized by Beck et al (2003a), even if associated with countries' initial endowments. While La Porta et al (1997) state that common law countries are more favourable to financial development, we find that German and French laws lead to higher ratios of loans to GDP. This conundrum is examined in paragraph 5.3.

Moving on to the control variables, per capita income has a positive impact on the ratio of loans to GDP. This is true using contemporaneous per capita income (regression 2 and 4) while the coefficients are positive but not significant using lagged per capita income in fixed and random effect models (regressions 3 and 5). When legal origin dummies are added to the random effect model (regression 6), we obtain similar results.

The econometric exercises of Table 5 also show that contemporaneous stock market capitalization has a positive but not statistically significant influence on the loans to GDP ratio, while predetermined stock market capitalization has a positive and significant coefficient in the random effect model (regression 5). Results are confirmed when legal origin dummies are added to the random effect and pooled models (regressions 6 and 7). This positive nexus confirms the hypothesis of complementarity between stock market capitalization and private credit highlighted in Levine and Zervos (1998).

Now we turn to some robustness checks of our econometric exercises.

#### 5. Robustness checks

#### 5.1 Do annual data regressions replicate the five-year average statistics results?

To increase the efficiency of our estimates, we run a further nine regressions by exploiting annual statistics (Table 6). We consider the same seven regressions of Table 5 plus two other exercises that include yearly time dummies. The estimated parameters confirm the previous results. Government debt maintains its negative and significant influence on the ratio of loans to GDP. A move from the sample mean of 62 per cent to a ratio of 52 per cent is associated with a 3.5 (regressions 2 and 3), or 5 (regressions 8 and 9) percentage points

increase in the loans to GDP ratio. Current inflation, its volatility, and the reserve requirement do not influence bank credit. The legal tradition confirms its role in explaining differences in the variability of banking business across countries: compared with common law countries, German and French legal origins are associated with higher ratios of loans to GDP. The German legal tradition dummy has the largest coefficients. Per capita GDP and stock market capitalization maintain a positive and often significant impact on the ratio of outstanding loans to GDP.

#### 5.2 Splitting the period into two sub-periods

Banking deregulation and liberalization were much stronger in the 1990s than in the 1980s. European directives liberalized banking markets but the same process also took place in other OECD countries such as the US.<sup>6</sup> There were also changes in inflation, the reserve requirement, government debt and capital account rules moving from the 1980s to the 1990s. For this reason we split our sample into two periods, 1981-1989 and 1990-1997 to search for possible breaks in the determinants of the ratio of credit to GDP. The results, given in Tables 7 and 8, confirm our previous results. Government debt has a negative influence on private credit and the positive sign of German legal origin is confirmed. As expected, the effect of the government debt to GDP ratio is stronger when the analysis is carried out over the period 1981-1989, as shown by the larger coefficients. In this period, countries which reduce the government debt to GDP ratio by 10 percentage points, are expected to raise credit to GDP ratio by 5.5 percentage points (see regressions 6 and 7) while for the period 1990-1997, the same decrease in government debt is associated with a rise in the dependent variable by 4 percentage points.

#### 5.3 Creditor rights vs. legal origin

In the previous paragraphs we found that the law and finance approach provides some insights to explain the size of bank loans in OECD countries. We now scrutinize this effect more closely, looking at the degree of protection of creditors as a determinant of banking development. In their aggregated indebtedness regressions over 39 countries, La Porta et al (1997) found significance for a creditor rights term. This indebtedness measure includes private sector bank debt and outstanding non-financial bonds. As in the previous paragraphs

<sup>&</sup>lt;sup>6</sup> On the transformation of the European banking system see Gual (1999), Belaisch et al (2001), Dermine (2000) and (2003). On the American banking system see Berger et al (1995) and Bernard and Bisignano (2006).

we concentrate on the determinants of bank loans because firms' bond issues respond to different motivations from the raising of bank credit (see paragraph 5.4 on firms' securities issued). Djankov et al (2007) found that stronger legal rights for creditors are associated with a higher level of private credit, running cross-section regressions for a set of 129 countries. On this basis, we investigate the linkage between creditor rights and the ratio of loans to GDP in our sample of OECD countries.

The creditor rights index has a positive effect on the ratio of bank loans to GDP (Table 9) but significance is only obtained when the legal origin dummies are left out of the model, probably because of a positive correlation between creditor rights and the German legal origin dummy. The signs and the significance obtained in the previous paragraph for the other regressors also remain unchanged using five-year average statistics (the results are not given here).

To sum up, we find that the German legal tradition leads to larger banking systems than the Anglo-Saxon and Scandinavian ones. Our conclusion is different from the typical results of the law and finance theory, which links the Anglo-Saxon legal origin to greater financial systems. La Porta et al (1997) stated that:

"[...] indebtedness is even higher in the German civil law countries – also sometimes described with bank-focused financial systems. [...] low rights line up with small markets when we compare French and English origin but German civil law countries are somewhat of a mystery.<sup>7</sup>

It is possible to reconcile our results with the law and finance view by observing that while we look at the determinants of bank credit, La Porta et al study the determinants of different notions of finance, i.e. stock market capitalization or bank loans plus bonds issued by firms. Our empirical results show that there is no German mystery: the German legal origin leads to more bank loans, which are an important source of external finance for firms. This evidence is compatible with Gerschenkron's (1962) thesis on the importance of the universal banks, not the stock market, for the economic growth of Germany and the other countries where this model of intermediary has spread.

<sup>&</sup>lt;sup>7</sup> According to the same authors, a possible explanation is suggested by Rajan and Zingales (1995), who find that German companies have high overall liabilities, though not necessarily high debt per se.

#### 5.4 Do results survive if corporate bonds are included in the picture ?

The size of bank loans might be influenced by firms' bond issues. One might argue that bank credit is more important where firms' securities issues are smaller (see Fink et al, 2005). To study the relevance of corporate bonds, we included in the model the ratio to GDP of bonds issued by non-financial corporations. For 13 OECD countries data on corporate bonds are available in a time-series format. The econometric exercise (Table 10) shows that the corporate securities coefficient is never significant. The introduction of firms' bonds does not affect the previous determinants of the ratio of credit to GDP.

#### 5.5 Other robustness checks

Here we discuss another four robustness checks of our regressions (for the sake of brevity the results are not given here but are available upon request).

First, Perotti and Von Thadden (2005) argue that in many countries the middle classes experienced the post-World War I inflationary shocks and their wealth endowments were seriously damaged; by consequence these classes turned toward more protected bank-oriented systems. Therefore, one should find a positive link between past inflation and the current weight of banks. We built an indicator of the post-World War I inflationary shocks, by averaging the annual growth rates of the consumer price index over the period 1910-1950 in some countries where historical statistics are available (see Mitchell, 2003). Inflationary shocks are statistically significant when there are no other independent variables. When we introduce other regressors - such as government debt and legal origin – past inflationary shocks lose their power to explain the current ratio of loans to GDP. We were not able to find a positive link between past inflationary shocks and current banking business.

Second, as seen in the review of the literature, some authors claim a positive connection between trade openness and banking development. Our results show that in most cases trade openness has a negative effect on bank loans but the coefficients are rarely statistically significant. Our findings do not support Rajan and Zingales's (2003) or Quy-Toan Do and Levchenko's (2004) views that there is positive linkage between trade and the size of financial systems.

Third, in the last few years some scholars improved the measurement of bank regulation and supervision indicators in order to understand national characteristics of banking business (see Barth et al, 2006). Exploiting a dataset created by the World Bank for many countries, these authors analyse how banking intermediation is influenced by measures such as entry barriers and competition regulation variables, restrictions on banking business, separation between banking and commerce, capital ratios, deposit insurance, official supervisory action variables, market discipline and transparency. This line of research is very interesting, but information on banking regulation and supervision is available in the form of cross-section statistics and refers only to very recent periods, while in this paper we study the development of banking loans over two decades. A harmonized picture of banking regulation is not available for the 1980s. For these reasons, the World Bank cross-section database is not used in our paper, even if we are aware that regulation is an important determinant of banking intermediation. Moreover, some variables used in the banking regulation approach, e.g. branching restrictions, are common to the financial repression theory. Pagano and Volpin (2001) claim that banking regulation is an example of where political factors are important, quoting the case of branching limits. Branching may be interpreted as an instrument to repress the geographical reach of banks. We use the ratio of branches to population in 1980 or 1985 as a determinant of the ratio of credit to GDP; we consider the lagged values of branches to avoid endogeneity issues with the level of loans. In our regressions, banking branches in 1980 (or 1985) are not significant to explain the ratio of loans to GDP in the years 1981-1997.

Fourth, we verified if the negative link between government debt and private credit survives after controlling for the importance of the State in the ownership of banks. The test consists of two regressions and makes use of statistics reported by La Porta et al (2002) for the years 1970 and 1995.<sup>8</sup> In the first test, a term capturing State ownership of banks is included as a determinant of private credit in an OLS equation running over a cross-section of statistics for 1995.<sup>9</sup> In the second test, a time-invariant term on State ownership of banks, as observed in 1970, is included as a determinant of private credit in a set of models for panel data (random effect, OLS with time-dummies, pooled regression). All the regressions show that government debt maintains a negative and significant linkage with private credit, even controlling for State ownership of banks. The signs and significance of the parameters mimic those estimated in the analysis of the previous sections. The coefficient for State ownership of banks is mainly negative but not statistically significant and does not affect the sign and the significance of government debt.

<sup>&</sup>lt;sup>8</sup> Other statistics on State ownership of banks are provided by Barth, Caprio and Levine on the World Bank website, but refer to data collected from 2001 onwards.

<sup>&</sup>lt;sup>9</sup> Outcomes provided by OLS regressions exploiting a few, i.e. 17, observations referring to OECD countries, are reported by Bleaney (1996).

#### 6. Conclusions

Our results show that two driving factors lie behind the differences in the size of bank credit relative to GDP.

First, government securities issues reduce the size of bank loans. A first explanation derives from a public finance perspective: financially repressed economies artificially increase their demand for State securities. Banks find that investing in government bonds is more attractive than granting credit to the economy. According to a second explanation, that does not exclude the first, a high public debt to GDP ratio may correspond to a large weight of the government and connected State-owned enterprises in the economy. In this setting, greater portions of credit are absorbed by governmental institutions and State-owned firms rather than by the private sector. While public debt has a negative impact on bank loans, other financial repression variables, such as inflation, its volatility and the reserve requirement do not show a robust significant effect on the ratio of loans to GDP.

Second, as far as the law and finance theory is concerned, we find that the legal tradition of countries is a significant factor that promotes bank credit. The object of our paper is not the study of the financial systems overall but only of the size of bank loans. Therefore we do not identify the Anglo-Saxon legal origin as the crucial driving variable for the development of credit systems. According to our evidence, the most powerful legal origin is the German one; such a result is compatible with the idea that the German legal system is nearer to the common law tradition than previously stated (see Mayer and Sussman, 2001).

Turning to the control variables, bank credit is often positively linked to stock market capitalization and per capita income, both contemporaneous and in lagged terms. Our empirical results are robust to the use of five-year average statistics or annual data, and to the introduction of firms' securities issued in the model. Finally, we did not find a significant effect of past inflationary shocks, the ratio of branches to population, and trade openness on the ratio of loans to GDP.

#### **Tables and Figures**

#### Table 1 - Data Definition and Sources

#### Sources of the statistics (key to abbreviations)

BDL: Beck, Demirgüç -Kunt, and Levine (2003b).
IFS: IMF, International Financial Statistics.
LLSV: La Porta, Lopez-de-Silanes, Shleifer, and Vishny (1997).
WDI: World Bank, World Development Indicators.
BP: OECD, Bank Profitability.
SC: OECD, Statistical Compendium.

#### Dependent variable

priv: claims on private sector by deposit money banks as a percentage of GDP, BDL.

#### Independent variables

res2:	reserve requirement calculated as ratio of domestic currency holdings
	and deposits with the monetary authorities to claims on other
	governments, non financial public enterprises, the private sector, and
	other banking institutions, WDI.
infl:	inflation rate calculated as annual growth rate of CPI, WDI.
sdinfl:	standard deviation of monthly inflation observations for the year, WDI.
gov_deb:	gross government debt as a percentage of GDP, SC.
legor_ge:	German legal origin, LLSV.
legor _fr:	French legal origin, LLSV.
legor_uk:	Anglo-Saxon legal origin, LLSV.
legor_sc:	Scandinavian legal origin of country, LLSV.
creditor rights:	an index aggregating creditor rights, LLSV. A score of one is assigned
	when each of the following rights of secured lenders are defined in laws
	and regulation: there are restrictions, such as creditor consent, for a
	debtor to file for reorganisation (i); secured creditors are able to seize
	their collateral once the reorganisation petition is approved (ii); secured
	creditors are paid first out of the proceeds of liquidating a bankrupt
	firm, as opposed to other creditors (iii); if management does not retain

	administration of its property pending the resolution of the										
	reorganisation (iv). The index ranges from 0 to 4.										
trade:	sum of exports and imports of goods and services measured a										
	percentage of GDP, WDI.										
infl_hist:	historical inflationary shocks, computed as averages of annual changes										
	of CPI, over the period 1910-1950. Source: Mitchell (2003).										
GDPpc:	per capita GDP (PPP, current international \$), WDI.										
stock:	stock market capitalisation as a percentage of GDP, BDL.										
corpbonds:	securities other than shares as a percentage of GDP, issued by non										
	financial corporations, European Central Bank and national financial										
	accounts.										

## Table 2 - Country list(18 High Income OECD Countries)

1	Australia
2	Austria
3	Belgium
4	Canada
5	Finland
6	France
7	Germany
8	Japan
9	Greece
10	Italy
11	Norway
12	Netherlands
13	Portugal
14	Spain
15	Sweden
16	Switzerland
17	United Kingdom
17 18	United Kingdom United States

#### **Table 3 - Descriptive statistics**

Annual data for the period 1981-1997. Priv is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); infl is the annual change in consumer prices; sdinfl is the standard deviation of monthly inflation observations for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the percapita GDP; stock is the stock market capitalization (as a percentage of GDP); trade is the sum of exports and imports of goods and services (as a percentage of GDP); corpbonds is the ratio of bonds issued by non financial corporations to GDP.

	priv	infl	sdinfl	res2	gov_deb	GDPpc	stock	trade	corpbonds
mean	71	0.05	0.01	5.0	62	16,564	37	59	11
sd	30	0.05	0.01	8.1	29	5,043	31	29	9
min	26	-0.19	0.00	0.2	12	6,337	0	16	1
max	167	0.34	0.06	81.7	144	30,123	183	149	30

#### Table 4 - Correlation matrix

Annual data for the period 1981-1997. Priv is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); infl is the annual change in consumer prices; sdinfl is the standard deviation of monthly inflation observations for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the percapita GDP; stock is the stock market capitalisation (as a percentage of GDP); trade is the sum of exports and imports of goods and services (as a share of GDP); corpbonds is the ratio of bonds issued by non financial corporations to GDP; legal dummies are denoted by the prefix legor\_.

	priv	infl	sdinfl	res2	gov_deb	GDPpc	stock	trade c	orpbonds	legor_ge	legor_fr	legor_uk	legor_sc
priv	1												
infl	-0.31	1											
sdinfl	-0.15	0.61	1										
res2	-0.26	0.49	0.34	1									
gov_deb	-0.36	-0.14	-0.12	0.02	1								
GDPpc	0.42	-0.60	-0.49	-0.49	0.15	1							
stock	0.50	-0.39	-0.27	-0.39	-0.03	0.59	1						
trade	-0.11	-0.08	-0.07	-0.15	0.36	0.02	-0.11	1					
corpbonds	0.01	-0.12	0.08	-0.25	0.01	0.46	0.54	-0.42	1				
legor_ge	0.68	-0.30	-0.17	-0.19	-0.21	0.22	0.16	-0.12	0.67	1			
legor_fr	-0.24	0.33	0.19	0.45	0.45	-0.36	-0.40	0.31	-0.24	-0.43	1		
legor_uk	-0.16	-0.07	-0.03	-0.21	-0.08	0.16	0.42	-0.32	-0.15	-0.29	-0.43	1	
legor_sc	-0.27	-0.02	-0.03	-0.14	-0.27	0.05	-0.12	0.07	-0.27	-0.24	-0.36	-0.24	1

#### Table 5 - Determinants of loans to GDP ratio in OECD countries: five-year average data

Five-year averages for the period 81-85, 86-90, 91-95, 96-00. Dependent variable (priv) is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); linfl is the logarithm of the annual change in consumer prices; Insdinfl is the logarithm of the standard deviation of monthly inflation observations for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the per capita GDP; stock is the stock market capitalisation (as a percentage of GDP); legal dummies are denoted by the prefix leg\_; the suffix (\_1) denotes the lagged term of variables; (\*) indicates marginal significance at the 5%-level, and (\*\*) at the 1% -level; FE and RE indicate fixed effect and random effect models. First diff. is the first difference estimator. (+) point estimates of legal dummies denote differential effects with respect to leg\_uk. Robust standard errors are below the estimated parameters.

	1	2	3	4	5	6	7
	FE	FE	FE	First diff.	RE	RE+	OLS+
linfl	-9.075 4.95	<b>1.966</b> 6.308	<b>3.322</b> 4.138	-0.430 3.899	<b>-0.291</b> 5.105	<b>1.511</b> 4.406	<b>1.595</b> 4.332
Insdinfl	-0.162 7.740	<b>8.612</b> 4.621	<b>6.291</b> <sup>5.279</sup>	9.010* 3.966	<b>2.151</b> 5.531	<b>2.556</b> 6.303	-4.783 6.095
res2	-0.215 <sub>0.225</sub>	<b>-0.170</b> <sub>0.134</sub>	-0.400 <sub>0.295</sub>	-0.188 0.330	<b>-0.423*</b> 0.204	<b>-0.680*</b> 0.288	-0.629 <sub>0.436</sub>
gov_deb		<b>-0.323*</b> 0.128	-0.278 0.150	<b>-0.388*</b> 0.162	<b>-0.364*</b> 0.152	-0.444** 0.070	-0.482** 0.070
GDPpc		<b>0.003**</b> 0.001		<b>0.003**</b> 0.000			
stock		<b>0.114</b> 0.120		0.012 0.086			
GDPpc_1			0.002 0.001		<b>0.001</b>	<b>0.001</b>	<b>0.001</b>
stock_1			0.206 <sub>0.144</sub>		<b>0.283*</b> 0.130	<b>0.408*</b> 0.161	<b>0.555**</b> 0.148
legor_ge						48.903** 9.176	49.002** 7.798
legor_fr						29.915** 8.286	33.334** 5.898
legor_sc						-4.115 10.467	<b>-0.348</b> 9.511
Constant	<b>41.978</b> 33.414	88.273** 20.973	108.132** 27.347		84.335** 26.506	<b>73.606*</b> 36.215	38.102 38.177
Ν	70	68	52	48	52	52	52
R-squared	0.21	0.30	0.34	0.27	0.43	0.85	0.86

#### Table 6 - Determinants of loans to GDP ratio in OECD countries

(annual data)

Annual data for the period 1981-1997. Dependent variable (priv) is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); linfl is the logarithm of the annual change in consumer prices; Insdinfl is the logarithm of the standard deviation of monthly inflation observation for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the per capita GDP; stock is the stock market capitalisation (as a percentage of GDP); legal dummies are denoted by the prefix leg\_; the suffix (\_1) denotes the lagged term of variables; (\*) indicates marginal significance at the 5%-level, and (\*\*) at the 1% level; FE and RE indicate fixed effect and random effect models. First diff. is the first difference estimator. (+) point estimates of legal dummies denote differential effects with respect to leg\_uk. Robust standard errors are below the estimated parameters.

	1	2	3	4	5	6	7	8	9
	FE	FE	FE	First diff.	RE	RE+	OLS+	FE ( time- dummies)	OLS (time dummies) +
linfl	-5.022** 1.229	<b>-0.366</b> 1.298	-0.537 1.302	<b>0.223</b> 0.500	-0.924 1.261	<b>-1.334</b> 1.256	<b>-3.39</b> 2.125	<b>0.544</b> 1.63	-1.992 3.108
Insdinfl	<b>-1.338</b> 1.903	<b>2.425</b> 1.590	<b>1.516</b> 1.463	0.075 <sub>0.356</sub>	<b>1.426</b> 1.495	<b>1.322</b> 1.499	<b>0.174</b>	<b>2.765</b> 1.696	1.043 <sup>2.074</sup>
res2	-0.879 <sub>0.427</sub>	-0.192 0.198	<b>-0.141</b> 0.183	<b>-0.308</b> 0.176	-0.106 0.185	-0.167 0.232	<b>0.027</b> 0.294	-0.05 <sub>0.245</sub>	-0.093 0.352
gov_deb		-0.357** 0.113	-0.341** 0.097	<b>-0.127*</b> 0.057	-0.356** 0.091	-0.411** 0.064	-0.494** 0.055	<b>-0.473</b> * 0.234	<b>-0.494**</b> 0.058
GDPpc		0.002** 0.000		<b>0.001**</b> 0.000				<b>0.001</b> 0.003	<b>0.001</b> 0.001
stock		<b>0.146</b> <sub>0.132</sub>		0.013 0.038				<b>0.164</b> 0.126	<b>0.41**</b> 0.410
GDPpc_1			0.002** 0.001		0.002** 0.001	0.002** 0.001	<b>0.001*</b> 0.000		
stock_1			0.179 <sub>0.160</sub>		<b>0.202</b> 0.142	<b>0.282*</b> 0.142	<b>0.441**</b> 0.101		
legor_ge						44.600** 6.587	43.460** 4.962		44.649** 4.648
legor_fr						27.004** 4.655	31.524** 3.167		30.440** 3.798
legor_sc						-2.680 <sup>7.092</sup>	-0.253 6.248		-1.124 6.379
Constant	50.584** 10.463	60.966** 10.37	58.607** 10.642		58.305** 11.852	<b>39.773**</b> 11.083	<b>31.982*</b> 12.896	<b>82.509*</b> 39.039	<b>40.029*</b> 16.903
N	291	277	262	253	262	262	262	277	277
R-squared	0.15	0.39	0.42	0.32	0.43	0.80	0.82	0.40	0.83

## Table 7 - Determinants of loans to GDP ratio in OECD countries (annual data, 1981-1989)

Annual data for the period 1981-1989. Dependent variable (priv) is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); linfl is the logarithm of the annual change in consumer prices; Insdinfl is the logarithm of the standard deviation of monthly inflation observations for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the per capita GDP; stock is the stock market capitalisation (as a percentage of GDP); legal dummies are denoted by the prefix leg\_; the suffix (\_1) denotes the lagged term of variables; (\*) indicates marginal significance at the 5%-level, and (\*\*) at the 1% - level; FE and RE indicate fixed effect and random effect models. First diff. is the first difference estimator. (+) point estimates of legal dummies denote differential effects with respect to leg\_uk. Robust standard errors are below the estimated parameters.

	1	2	3	4	5	6	7	8	9
	FE	FE	FE	First diff.	RE	RE+	OLS+	FE ( time- dummies)	OLS (time dummies)+
linfl	<b>-2.452</b> 1.331	<b>-1.811</b> 1.081	<b>-1.589</b> 1.767	-0.419 <sub>0.845</sub>	-2.215 1.808	<b>-1.096</b> 1.722	-5.234 2.759	<b>1.050</b> 1.96	-4.506 <sup>2.823</sup>
Insdinfl	<b>-2.397</b> 1.518	<b>1.798</b> 1.097	<b>1.031</b> 1.372	<b>0.342</b> <sub>0.447</sub>	<b>2.026</b> 1.613	<b>1.568</b> 1.621	<b>2.313</b> 2.521	<b>1.392</b> 1.023	<b>2.671</b> 2.776
res2	<b>-2.040**</b> 0.441	-0.987** 0.235	-1.150** 0.197	-0.176 <sub>0.240</sub>	-0.346 <sub>0.251</sub>	-0.483** 0.138	<b>0.462*</b> <sub>0.211</sub>	-1.080** 0.200	<b>0.702*</b> 0.295
gov_deb		-0.600** 0.206	<b>-0.584*</b> 0.208	-0.333** 0.099	-0.515** 0.125	-0.545** 0.106	-0.563** 0.065	<b>-0.726*</b> 0.315	-0.588** 0.071
GDPpc		<b>0.002*</b> -0.001		<b>0.001**</b> 0.000				0.003 0.002	<b>0.002**</b> 0.001
stock		<b>0.234</b> 0.162		0.230** 0.078				<b>0.196</b> <sub>0.149</sub>	<b>0.445**</b> 0.097
GDPpc_1			<b>0.002*</b> 0.001		0.002** 0.000	0.002** 0.001	<b>0.001*</b> 0.000		
stock_1			0.307 <sub>0.233</sub>		<b>0.315*</b> 0.150	<b>0.38</b> 0.201	<b>0.503**</b> 0.086		
legor_ge						48.541** 7.216	41.837** 4.556		<b>41.321**</b> 4.103
legor_fr						<b>36.313**</b> 9.600	28.543** 5.585		29.362** 5.550
legor_sc						3.102 8.919	<b>2.597</b> 4.662		1.139 <sup>4.847</sup>
Constant	53.714** 7.165	70.456** 6.607	71.018** 7.432		67.248** 14.308	<b>45.296**</b> 13.064	<b>39.440*</b> 17.936	<b>73.640*</b> 32.096	<b>20.878</b> 21.871
Ν	151	137	122	116	122	122	122	137	137
R-squared	0.02	0.25	0.26	0.29	0.33	0.78	0.83	0.64	0.83

## Table 8 - Determinants of loans to GDP ratio in OECD countries (annual data, 1990-1997)

Annual data for the period 1990-1997. Dependent variable (priv) is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); linfl is the logarithm of the annual change in consumer prices; Insdinfl is the logarithm of the standard deviation of monthly inflation observations for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the per capita GDP; stock is the stock market capitalisation (as a percentage of GDP); legal dummies are denoted by the prefix leg\_; the suffix (\_1) denotes the lagged term of variables; (\*) indicates marginal significance at the 5%-level, and (\*\*) at the 1% -level; FE and RE indicate fixed effect and random effect models. First diff. is the first difference estimator. (+) point estimates of legal dummies denote differential effects with respect to leg\_uk. Robust standard errors are below the estimated parameters.

	1	2	3	4	5	6	7	8	9
	FE	FE	FE	First diff.	RE	RE+	OLS+	FE ( time- dummies)	OLS (time dummies)+
linfl	<b>0.793</b> 1.522	-0.357 0.685	-0.201 0.683	0.192 <sub>0.633</sub>	-0.363 <sub>0.744</sub>	-0.737 <sub>0.729</sub>	-3.742 2.600	<b>1.984</b>	-3.250 3.761
Insdinfl	<b>-1.281</b> 0.852	<b>-1.930</b>	<b>-2.025*</b> 0.847	<b>-0.645</b> 0.601	<b>-2.059*</b> 0.848	-1.770* <sub>0.786</sub>	<b>-1.389</b> 1.983	-0.972 0.992	-0.008 2.359
res2	<b>-0.841*</b> 0.344	-0.625* <sub>0.281</sub>	<b>-0.536</b> <sub>0.313</sub>	<b>-0.399</b> 0.261	<b>-0.583*</b> 0.280	-0.649** 0.241	-0.818 <sub>0.423</sub>	-0.009 <sub>0.293</sub>	<b>-0.985</b> <sub>0.512</sub>
gov_deb		-0.331** 0.104	-0.361** 0.108	-0.011 0.074	-0.380** 0.110	-0.402** 0.090	-0.441** 0.089	-0.771** 0.196	-0.415** 0.088
GDPpc		<b>0.002</b> -0.001		<b>0.000</b> 0.000				-0.004 0.002	<b>0.000</b> 0.001
stock		-0.047 <sub>0.07</sub>		-0.018 0.047				-0.143 0.08	<b>0.406**</b> 0.094
GDPpc_1			<b>0.002</b> 0.001		<b>0.002</b> 0.001	<b>0.002</b> 0.001	<b>0.000</b> 0.001		
stock_1			-0.082 0.095		-0.038 0.090	<b>0.011</b> 0.080	<b>0.409**</b> 0.109		
legor_ge						<b>38.313**</b> 14.722	<b>44.747**</b> 7.491		<b>47.603**</b> 6.795
legor_fr						<b>16.238</b> 11.321	30.205** 6.687		30.301** 6.835
legor_sc						-15.349 13.295	-2.627 10.758		<b>-1.754</b> 11.554
Constant	77.067** 6.103	63.061** 16.988	59.464** 16.779		59.124** 16.062	<b>49.997*</b> 21.397	<b>45.204</b> 23.539	205.922** 52.897	<b>50.107</b> 31.736
Ν	140	140	140	119	140	140	140	140	140
R-squared	0.18	0.34	0.33	0.18	0.37	0.72	0.83	0.02	0.84

## Table 9 - Determinants of loans to GDP ratio in OECD countries: do creditor rights matter ?

#### (annual data)

Annual data for the period 1981-1997. Dependent variable (priv) is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); linfl is the logarithm of the annual change in consumer prices; Insdinfl is the logarithm of the standard deviation of monthly inflation observations for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the per capita GDP; stock is the stock market capitalisation (as a percentage of GDP); or\_rights is an index indicating creditor rights; legal dummies are denoted by the prefix leg\_; the suffix (\_1) denotes the lagged term of variables; (\*) indicates marginal significance at the 5%-level, and (\*\*) at the 1% -level; RE indicates random effect model. (+) point estimates of legal dummies denote differential effects with respect to leg\_uk. Robust standard errors are below the estimated parameters.

	1	2	3
	RE+	RE+	RE
linfl	<b>-1.734</b> 1.294	<b>-1.215</b> 1.252	<b>-0.686</b> 1.283
Insdinfl	<b>1.073</b> 1.133		
res2	-0.159 <sub>0.216</sub>	-0.137 0.225	-0.014 <sub>0.241</sub>
gov_deb	-0.379** 0.051	-0.377** 0.054	-0.318** 0.057
GDPpc_1	0.002** 0.000	<b>0.002**</b> 0.000	<b>0.002**</b> 0.000
stock_1	0.280** 0.055	0.253** 0.058	0.198** 0.062
cr_rights		<b>3.471</b> 2.460	<b>7.464</b> * 3.223
legor_ge	<b>39.482**</b> 5.434	<b>35.683**</b> 6.960	
legor_fr	25.679** 5.034	25.777** <sup>5.936</sup>	
legor_sc	<b>-2.522</b> 5.543	-1.893 6.690	
Constant	38.397** 6.900	28.446** 7.444	36.266** 7.779
Ν	251	251	251
R-squared	0.69	0.70	0.35

## Table 10 - Determinants of loans to GDP ratio in OECD countries: do corporate bonds matter ? (annual data)

Annual data for the period 1981-1997. Dependent variable (priv) is the amount of credit granted by the bank system to the private sector (as a percentage of GDP); linfl is the logarithm of the annual change in consumer prices; Insdinfl is the logarithm of the standard deviation of monthly inflation observations for the year; res2 is the bank liquid reserves to bank assets ratio; gov\_deb is the gross government debt (as a percentage of GDP); GDPpc is the per capita GDP; stock is the stock market capitalisation (as a percentage of GDP); corpbond is the stock of bonds issued by non financial corporations (as a percentage of GDP): legal dummies are denoted by the prefix leg\_; the suffix (\_1) denotes the lagged term of variables; (\*) indicates marginal significance at the 5%-level, and (\*\*) at the 1% -level; FE and RE indicate fixed effect and random effect models. First diff. is the first difference estimator. (+) point estimates of legal dummies denote differential effects with respect to leg\_uk. Robust standard errors are below the estimated parameters.

	1	2	3	4	5	6	7	8	9
	FE	FE	FE	First diff.	RE	RE+	OLS+	FE ( time- dummies)	OLS (time dummies)+
linfl	-0.429 1.964	-0.932 <sub>0.764</sub>	-1.042 <sup>0.979</sup>	0.029 <sub>0.830</sub>	-1.707 1.045	<b>-2.947*</b> 1.373	-4.045 2.737	<b>1.270</b> 2.054	-0.586 1.926
Insdinfl	-0.678 0.950	-0.510 <sub>0.903</sub>	-0.427 <sub>0.799</sub>	-0.548 <sub>0.629</sub>	<b>-0.494</b> 0.872	-0.148	<b>0.639</b> 1.360	-0.729	<b>1.002</b> 1.906
res2	-0.713** 0.130	-0.473 <sub>0.233</sub>	-0.416 <sub>0.248</sub>	-0.448 0.301	<b>-0.498*</b> 0.247	<b>-0.673*</b> 0.331	-0.916* <sub>0.417</sub>	0.293 <sub>0.403</sub>	<b>-1.100*</b> 0.423
gov_deb		-0.268* 0.111	<b>-0.293*</b> 0.108	0.017 0.092	-0.344** 0.074	-0.388** 0.075	-0.348** 0.058	-0.806** 0.241	-0.331** 0.068
GDPpc		<b>0.002</b> 0.001		<b>0.000</b> 0.001				-0.007 0.004	-0.001 0.001
stock		-0.058 <sub>0.106</sub>		-0.034 <sub>0.057</sub>				-0.119 <sub>0.104</sub>	<b>0.367**</b> 0.073
GDPpc_1			0.002 0.001		0.002 0.001	<b>0.001</b>	0.000 0.001		
stock_1			-0.067 0.144		-0.034 0.122	<b>0.139</b> 0.107	0.382** 0.077		
corpbonds	<b>1.616</b> 0.886	<b>1.236</b> 0.811	<b>1.179</b> 0.762	<b>1.357</b> 0.385	<b>0.428</b> 0.611	<b>0.065</b> 0.545	-0.498 0.309	<b>0.768</b> 0.794	-0.279 0.309
legor_ge						28.939* 13.511	<b>30.084**</b> 5.129		34.923** 4.777
legor_fr						<b>18.632</b> 13.368	15.569* 5.935		16.418* 6.726
legor_sc						<b>4.692</b> 13.431	<b>6.358</b> 5.655		<b>9.424</b> 5.009
Constant	59.390** 12.475	<b>48.745</b> 23.185	<b>46.906</b> 22.990		57.695** 21.604	<b>54.668</b> 31.132	77.449** 24.378	<b>306.783*</b> 101.521	98.929** 29.776
Ν	115	115	115	100	115	115	115	115	115
R-squared	0.01	0.09	0.10	0.00	0.24	0.67	0.78	0.07	0.80

#### References

- Alcalá, F. and A. Ciccone (2003), "Trade, Extent of the Market, and Economic Growth 1960-1996," *Universitat Pompeu Fabra*, Working Papers, No 765.
- Allen, F., L. Bartiloro and O. Kowalewski (2006), "Does Economic Structure Determine Financial Structure ?", mimeo.
- Barth, J. R., G. J. Caprio and R. Levine (2006), "*Rethinking Bank Regulation: Till Angels Govern*", Cambridge University Press.
- Battilossi, S. (2003), "Capital Mobility and Financial Repression in Italy, 1960-1990: A Public Finance Perspective", *Universidad Carlos III De Madrid*, Working Paper in Economic History, No 03-06.
- Beck, T., A. Demirgüç-Kunt and R. Levine R. (2003a), "Law, Endowments, and Finance", *Journal of Financial Economics*, Vol. 70(2), pp. 137-181.
- Beck, T., A. Demirgüç-Kunt and R. Levine (2003b), "A New Database on Financial Development and Structure", *World Bank Economic Review*, Vol 14, pp. 597-605 updated in 2008.
- Belaisch, A., L. Kodres, J. Levy and A. Ubide (2001), "Euro-Area Banking at the Crossroads", International Monetary Found, IMF Working Paper, No. 1/28.
- Berger, A.N., A.K. Kashyap and J.M. Scalise (1995), "The Transformation of the US Banking Industry: What Long, Strange Trip it's been", *Brookings Papers on Economic Activity*, Vol. 26 (2), pp. 55-218.
- Bernard, H. and J. Bisignano (2006), "Financial Intermediary Transformation. Risk Absorption, Transfer and Trading in the US Financial System", in M. de Cecco and G. Nardozzi (Eds.), *Banche e finanza nel futuro: Europa, Stati Uniti, Asia*, Bancaria Editrice.
- Bleaney, M. (1996), "Central Bank Independence, Wage Bargaining Structure, and Macroeconomic Performance in OECD Countries", *Oxford Economic Papers*, Vol 48, pp. 20-38.
- Boyd, J.H., R. Levine and B.D. Smith (2001), "The Impact of Inflation on Financial Sector Performance", *Journal of Monetary Economics*, Vol. 47(2), pp. 221-248.
- Brock, P. L. (1989), "Reserve Requirements and the Inflation Tax", *Journal of Money Credit and Banking*, Vol. 21(1), pp. 106-221.
- Chinn, M. D. and H. Ito (2006), "What Matters for Financial Development? Capital Controls, Institutions, and Interactions", *Journal of Development Economics*, Vol. 81(1), pp. 163-192.
- Demirgüç-Kunt, A. and R. Levine (2001), "Financial Structure and Economic Growth", The MIT Press.

- Dermine, J. (2000), "Eurobanking", in M. Artis, A. Weber and E. Hennessy (Eds.), *The Euro. A Challenge and Opportunity for Financial Markets*, Routledge International Studies in Money and Banking.
- Dermine, J. (2003), "Banking in Europe: Past, Present and Future", in V. Gaspar, P. Hartmann and O. Sleijpen (Eds.), *The Transformation of the European Financial System*, European Central Bank, Frankfurt.
- Djankov, S., C. McLiesh and A. Shleifer A. (2007), "Private Credit in 129 Countries", *Journal of Financial Economics*, Vol. 84(2), pp. 299-329.
- Fink, G., P. Haiss and S. Hristoforova (2005), "Credit, Bonds and Stocks in Seven Large Economies", available at <u>http://fgr.wu-wien.ac.at/institut/ef/nexus.html</u>.
- Gerschenkron, A. (1962), "Economic Backwardness in Historical Perspective", Harvard University Press.
- Giovannini, A. and M. De Melo (1991), "Government Revenue from Financial Repression", *American Economic Review*, Vol. 83, No. 4, pp. 953-963.
- Gual, J. (1999), "Deregulation, Integration and Market Structure in the European Banking", *CEPR Discussion Paper*, No. 2288.
- Haslag, J. and J. Koo (1999), "Financial Repression, Financial Development and Economic Growth", *Federal Reserve of Dallas*, Working Paper, No 99-02.
- Hauner, D. (2008), "Public debt and financial development", *Journal of Development Economics*, Vol. 88 (2), pp. 171–183.
- Islam, N. (1995), "Growth Empirics: A Panel Data Approach", *Quarterly Journal of Economics*, Vol. 110(4), pp. 1127-1170.
- La Porta, R., F. Lopez de Silanes, A. Shleifer and R W. Vishny (1997), "Legal Determinants of External Finance", *Journal of Finance*, Vol. 52(3), pp. 1131-50.
- La Porta, R., F. Lopez de Silanes F., and A. Shleifer (2002), "Government Ownership of Banks", *The Journal of Finance*, Vol. 72 (1), pp. 265-301.
- Levine, R. (2003), "More on Finance and Growth: More Finance, More Growth?", *Federal Reserve Bank of St. Louis Review*, Vol. 85, pp. 31–46.
- Levine, R. and S. Zervos (1998), "Stock Markets, Banks, and Economic Growth", American Economic Review, Vol. 88(3), pp. 537-558.
- Levine, R. N. Loayza and T. Beck (2000), "Financial Intermediation and Growth: Causality and Causes", *Journal of Monetary Economics*, Vol. 46(1), pp. 31-77.
- Mayer, C. and O. Sussman (2001), "The Assessment: Finance, Law and Growth", Oxford Review of Economic Policy, Vol. 17(4), pp. 457-466.

- McKinnon, R. (1973), "Money and Capital in Economic Development", Washington, DC, The Brooking Institution.
- Mitchell, B.R. (2003), "International Historical Statistics 1750-2005", New York, Palgrave Macmillan.
- Pagano, M. and P. Volpin (2001), "The Political Economy of Finance", Oxford Review of Economic Policy, Vol. 17(4), pp. 502-519.
- Perotti, E. and E.L. Von Thadden (2005), "The Political Economy of Corporate Control", mimeo.
- Piluso, G. and R. Ricciuti (2008), "Fiscal Policy and the Banking System in Italy. Have Taxes, Public Spending and Banks been Procyclical in the Long-Run?", *Cesifo Working Paper*, No. 2442.
- Quy-Toan, Do and A. Levchenko (2004), "Trade and Financial Development", *World Bank*, Policy, Research Working Paper Series, Vol. 3347.
- Rajan, R.G. and L. Zingales (1995), "What Do We Know From Capital Structure: Some Evidence From the International Data", *Journal of Finance*, Vol. 50 (5), pp. 1421-1460.
- Rajan, R.G. and L. Zingales (2003), "The Great Reversals: The Politics of Financial Development in the 20th Century", *Journal of Financial Economics*, Vol. 69(1), pp. 5-50.
- Roubini, N. and X. Sala-i-Martin (1995), "A growth model of inflation, tax evasion, and financial repression", *Journal of Monetary Economics*, Vol. 35(2), pp. 275-301.
- Shaw, E. (1973), "Financial Deepening in Economic Development", New York, Oxford University Press.

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 680 Family succession and firm performance: Evidence from Italian family firms, by Marco Cucculelli and Giacinto Micucci (June 2008).
- N. 681 Short-term interest rate futures as monetary policy forecasts, by Giuseppe Ferrero and Andrea Nobili (June 2008).
- N. 682 Vertical specialisation in Europe: Evidence from the import content of exports, by Emanuele Breda, Rita Cappariello and Roberta Zizza (August 2008).
- N. 683 A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, by Andrea Mercatanti (August 2008).
- N. 684 Balancing work and family in Italy: New mothers' employment decisions after childbirth, by Piero Casadio, Martina Lo Conte and Andrea Neri (August 2008).
- N. 685 *Temporal aggregation of univariate and multivariate time series models: A survey*, by Andrea Silvestrini and David Veredas (August 2008).
- N. 686 Exploring agent-based methods for the analysis of payment systems: A crisis model for StarLogo TNG, by Luca Arciero, Claudia Biancotti, Leandro D'Aurizio and Claudio Impenna (August 2008).
- N. 687 The labor market impact of immigration in Western Germany in the 1990's, by Francesco D'Amuri, Gianmarco I. P. Ottaviano and Giovanni Peri (August 2008).
- N. 688 Agglomeration and growth: the effects of commuting costs, by Antonio Accetturo (September 2008).
- N. 689 A beta based framework for (lower) bond risk premia, by Stefano Nobili and Gerardo Palazzo (September 2008).
- N. 690 *Nonlinearities in the dynamics of the euro area demand for M1*, by Alessandro Calza and Andrea Zaghini (September 2008).
- N. 691 Educational choices and the selection process before and after compulsory schooling, by Sauro Mocetti (September 2008).
- N. 692 Investors' risk attitude and risky behavior: a Bayesian approach with imperfect information, by Stefano Iezzi (September 2008).
- N. 693 Competing influence, by Enrico Sette (September 2008).
- N. 694 La relazione tra gettito tributario e quadro macroeconomico in Italia, by Alberto Locarno and Alessandra Staderini (December 2008).
- N. 695 *Immigrant earnings in the Italian labour market*, by Antonio Accetturo and Luigi Infante (December 2008).
- N. 696 *Reservation wages: explaining some puzzling regional patterns*, by Paolo Sestito and Eliana Viviano (December 2008).
- N. 697 Technological change and the demand for currency: an analysis with household data, by Francesco Lippi and Alessandro Secchi (December 2008).
- N. 698 *Immigration and crime: an empirical analysis*, by Milo Bianchi, Paolo Buonanno and Paolo Pinotti (December 2008).
- N. 699 Bond risk premia, macroeconomic fundamentals and the exchange rate, by Marcello Pericoli and Marco Taboga (January 2009).
- N. 700 What determines debt intolerance? The role of political and monetary institutions, by Raffaela Giordano and Pietro Tommasino (January 2009).
- N. 701 On analysing the world distribution of income, by Anthony B. Atkinson and Andrea Brandolini (January 2009).
- N. 702 Dropping the books and working off the books, by Rita Cappariello and Roberta Zizza (January 2009).
- N. 703 Measuring wealth mobility, by Andrea Neri (January 2009).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

#### 2006

- F. BUSETTI, Tests of seasonal integration and cointegration in multivariate unobserved component models, Journal of Applied Econometrics, Vol. 21, 4, pp. 419-438, **TD No. 476 (June 2003).**
- C. BIANCOTTI, A polarization of inequality? The distribution of national Gini coefficients 1970-1996, Journal of Economic Inequality, Vol. 4, 1, pp. 1-32, **TD No. 487 (March 2004).**
- L. CANNARI and S. CHIRI, *La bilancia dei pagamenti di parte corrente Nord-Sud (1998-2000)*, in L. Cannari, F. Panetta (a cura di), Il sistema finanziario e il Mezzogiorno: squilibri strutturali e divari finanziari, Bari, Cacucci, **TD No. 490 (March 2004)**.
- M. BOFONDI and G. GOBBI, *Information barriers to entry into credit markets*, Review of Finance, Vol. 10, 1, pp. 39-67, **TD No. 509 (July 2004).**
- W. FUCHS and LIPPI F., *Monetary union with voluntary participation*, Review of Economic Studies, Vol. 73, pp. 437-457 TD No. 512 (July 2004).
- E. GAIOTTI and A. SECCHI, *Is there a cost channel of monetary transmission? An investigation into the pricing behaviour of 2000 firms*, Journal of Money, Credit and Banking, Vol. 38, 8, pp. 2013-2038
   TD No. 525 (December 2004).
- A. BRANDOLINI, P. CIPOLLONE and E. VIVIANO, *Does the ILO definition capture all unemployment?*, Journal of the European Economic Association, Vol. 4, 1, pp. 153-179, **TD No. 529 (December 2004).**
- A. BRANDOLINI, L. CANNARI, G. D'ALESSIO and I. FAIELLA, *Household wealth distribution in Italy in the* 1990s, in E. N. Wolff (ed.) International Perspectives on Household Wealth, Cheltenham, Edward Elgar, **TD No. 530 (December 2004).**
- P. DEL GIOVANE and R. SABBATINI, Perceived and measured inflation after the launch of the Euro: Explaining the gap in Italy, Giornale degli economisti e annali di economia, Vol. 65, 2, pp. 155-192, TD No. 532 (December 2004).
- M. CARUSO, *Monetary policy impulses, local output and the transmission mechanism*, Giornale degli economisti e annali di economia, Vol. 65, 1, pp. 1-30, **TD No. 537 (December 2004).**
- L. GUISO and M. PAIELLA, The role of risk aversion in predicting individual behavior, In P. A. Chiappori e C. Gollier (eds.) Competitive Failures in Insurance Markets: Theory and Policy Implications, Monaco, CESifo, **TD No. 546 (February 2005).**
- G. M. TOMAT, Prices product differentiation and quality measurement: A comparison between hedonic and matched model methods, Research in Economics, Vol. 60, 1, pp. 54-68, TD No. 547 (February 2005).
- L. GUISO, M. PAIELLA and I. VISCO, *Do capital gains affect consumption? Estimates of wealth effects from Italian household's behavior*, in L. Klein (ed), Long Run Growth and Short Run Stabilization: Essays in Memory of Albert Ando (1929-2002), Cheltenham, Elgar, **TD No. 555 (June 2005).**
- F. BUSETTI, S. FABIANI and A. HARVEY, *Convergence of prices and rates of inflation*, Oxford Bulletin of Economics and Statistics, Vol. 68, 1, pp. 863-878, **TD No. 575 (February 2006).**
- M. CARUSO, Stock market fluctuations and money demand in Italy, 1913 2003, Economic Notes, Vol. 35, 1, pp. 1-47, **TD No. 576 (February 2006).**
- R. BRONZINI and G. DE BLASIO, *Evaluating the impact of investment incentives: The case of Italy's Law* 488/92. Journal of Urban Economics, Vol. 60, 2, pp. 327-349, **TD No. 582 (March 2006).**
- R. BRONZINI and G. DE BLASIO, *Una valutazione degli incentivi pubblici agli investimenti*, Rivista Italiana degli Economisti, Vol. 11, 3, pp. 331-362, **TD No. 582** (March 2006).
- A. DI CESARE, *Do market-based indicators anticipate rating agencies? Evidence for international banks*, Economic Notes, Vol. 35, pp. 121-150, **TD No. 593 (May 2006).**
- R. GOLINELLI and S. MOMIGLIANO, *Real-time determinants of fiscal policies in the euro area*, Journal of Policy Modeling, Vol. 28, 9, pp. 943-964, **TD No. 609 (December 2006).**

- S. SIVIERO and D. TERLIZZESE, *Macroeconomic forecasting: Debunking a few old wives' tales,* Journal of Business Cycle Measurement and Analysis, v. 3, 3, pp. 287-316, **TD No. 395 (February 2001).**
- S. MAGRI, Italian households' debt: The participation to the debt market and the size of the loan, Empirical Economics, v. 33, 3, pp. 401-426, **TD No. 454 (October 2002).**
- L. CASOLARO. and G. GOBBI, *Information technology and productivity changes in the banking industry*, Economic Notes, Vol. 36, 1, pp. 43-76, **TD No. 489 (March 2004).**
- G. FERRERO, *Monetary policy, learning and the speed of convergence*, Journal of Economic Dynamics and Control, v. 31, 9, pp. 3006-3041, **TD No. 499 (June 2004).**
- M. PAIELLA, Does wealth affect consumption? Evidence for Italy, Journal of Macroeconomics, Vol. 29, 1, pp. 189-205, TD No. 510 (July 2004).
- F. LIPPI. and S. NERI, *Information variables for monetary policy in a small structural model of the euro area*, Journal of Monetary Economics, Vol. 54, 4, pp. 1256-1270, **TD No. 511 (July 2004).**
- A. ANZUINI and A. LEVY, *Monetary policy shocks in the new EU members: A VAR approach*, Applied Economics, Vol. 39, 9, pp. 1147-1161, **TD No. 514 (July 2004).**
- D. JR. MARCHETTI and F. Nucci, *Pricing behavior and the response of hours to productivity shocks*, Journal of Money Credit and Banking, v. 39, 7, pp. 1587-1611, **TD No. 524 (December 2004).**
- R. BRONZINI, FDI Inflows, agglomeration and host country firms' size: Evidence from Italy, Regional Studies, Vol. 41, 7, pp. 963-978, TD No. 526 (December 2004).
- L. MONTEFORTE, Aggregation bias in macro models: Does it matter for the euro area?, Economic Modelling, 24, pp. 236-261, **TD No. 534 (December 2004).**
- A. NOBILI, Assessing the predictive power of financial spreads in the euro area: does parameters instability matter?, Empirical Economics, Vol. 31, 1, pp. 177-195, **TD No. 544 (February 2005).**
- A. DALMAZZO and G. DE BLASIO, *Production and consumption externalities of human capital: An empirical study for Italy*, Journal of Population Economics, Vol. 20, 2, pp. 359-382, **TD No. 554 (June 2005).**
- M. BUGAMELLI and R. TEDESCHI, Le strategie di prezzo delle imprese esportatrici italiane, Politica Economica, v. 23, 3, pp. 321-350, TD No. 563 (November 2005).
- L. GAMBACORTA and S. IANNOTTI, Are there asymmetries in the response of bank interest rates to monetary shocks?, Applied Economics, v. 39, 19, pp. 2503-2517, TD No. 566 (November 2005).
- P. ANGELINI and F. LIPPI, *Did prices really soar after the euro cash changeover? Evidence from ATM withdrawals*, International Journal of Central Banking, Vol. 3, 4, pp. 1-22, **TD No. 581 (March 2006).**
- A. LOCARNO, Imperfect knowledge, adaptive learning and the bias against activist monetary policies, International Journal of Central Banking, v. 3, 3, pp. 47-85, **TD No. 590 (May 2006).**
- F. LOTTI and J. MARCUCCI, *Revisiting the empirical evidence on firms' money demand*, Journal of Economics and Business, Vol. 59, 1, pp. 51-73, **TD No. 595 (May 2006).**
- P. CIPOLLONE and A. ROSOLIA, *Social interactions in high school: Lessons from an earthquake*, American Economic Review, Vol. 97, 3, pp. 948-965, **TD No. 596 (September 2006).**
- L. DEDOLA and S. NERI, *What does a technology shock do? A VAR analysis with model-based sign restrictions*, Journal of Monetary Economics, Vol. 54, 2, pp. 512-549, **TD No. 607 (December 2006).**
- F. VERGARA CAFFARELLI, *Merge and compete: strategic incentives for vertical integration*, Rivista di politica economica, v. 97, 9-10, serie 3, pp. 203-243, **TD No. 608 (December 2006).**
- A. BRANDOLINI, Measurement of income distribution in supranational entities: The case of the European Union, in S. P. Jenkins e J. Micklewright (eds.), Inequality and Poverty Re-examined, Oxford, Oxford University Press, TD No. 623 (April 2007).
- M. PAIELLA, *The foregone gains of incomplete portfolios*, Review of Financial Studies, Vol. 20, 5, pp. 1623-1646, **TD No. 625 (April 2007).**
- K. BEHRENS, A. R. LAMORGESE, G.I.P. OTTAVIANO and T. TABUCHI, Changes in transport and non transport costs: local vs. global impacts in a spatial network, Regional Science and Urban Economics, Vol. 37, 6, pp. 625-648, TD No. 628 (April 2007).
- M. BUGAMELLI, Prezzi delle esportazioni, qualità dei prodotti e caratteristiche di impresa: analisi su un campione di imprese italiane, v. 34, 3, pp. 71-103, Economia e Politica Industriale, TD No. 634 (June 2007).
- G. ASCARI and T. ROPELE, *Optimal monetary policy under low trend inflation*, Journal of Monetary Economics, v. 54, 8, pp. 2568-2583, **TD No. 647** (November 2007).
- R. GIORDANO, S. MOMIGLIANO, S. NERI and R. PEROTTI, *The Effects of Fiscal Policy in Italy: Evidence from a VAR Model*, European Journal of Political Economy, Vol. 23, 3, pp. 707-733, **TD No. 656** (January 2008).

- B. ROFFIA and A. ZAGHINI, *Excess money growth and inflation dynamics*, International Finance, v. 10, 3, pp. 241-280, **TD No. 657 (January 2008).**
- G. BARBIERI, P. CIPOLLONE and P. SESTITO, Labour market for teachers: demographic characteristics and allocative mechanisms, Giornale degli economisti e annali di economia, v. 66, 3, pp. 335-373, TD No. 672 (June 2008).
- E. BREDA, R. CAPPARIELLO and R. ZIZZA, Vertical specialisation in Europe: evidence from the import content of exports, Rivista di politica economica, numero monografico, TD No. 682 (August 2008).

2008

- P. ANGELINI, *Liquidity and announcement effects in the euro area*, Giornale degli Economisti e Annali di Economia, v. 67, 1, pp. 1-20, **TD No. 451 (October 2002).**
- P. ANGELINI, P. DEL GIOVANE, S. SIVIERO and D. TERLIZZESE, Monetary policy in a monetary union: What role for regional information?, International Journal of Central Banking, v. 4, 3, pp. 1-28, TD No. 457 (December 2002).
- F. SCHIVARDI and R. TORRINI, *Identifying the effects of firing restrictions through size-contingent Differences in regulation*, Labour Economics, v. 15, 3, pp. 482-511, **TD No. 504 (June 2004).**
- L. GUISO and M. PAIELLA,, *Risk aversion, wealth and background risk*, Journal of the European Economic Association, v. 6, 6, pp. 1109-1150, **TD No. 483 (September 2003).**
- C. BIANCOTTI, G. D'ALESSIO and A. NERI, *Measurement errors in the Bank of Italy's survey of household income and wealth*, Review of Income and Wealth, v. 54, 3, pp. 466-493, **TD No. 520 (October 2004).**
- S. MOMIGLIANO, J. HENRY and P. HERNÁNDEZ DE COS, The impact of government budget on prices: Evidence from macroeconometric models, Journal of Policy Modelling, v. 30, 1, pp. 123-143 TD No. 523 (October 2004).
- L. GAMBACORTA, *How do banks set interest rates?*, European Economic Review, v. 52, 5, pp. 792-819, **TD No. 542 (February 2005).**
- P. ANGELINI and A. GENERALE, *On the evolution of firm size distributions*, American Economic Review, v. 98, 1, pp. 426-438, **TD No. 549 (June 2005).**
- R. FELICI and M. PAGNINI, *Distance, bank heterogeneity and entry in local banking markets*, The Journal of Industrial Economics, v. 56, 3, pp. 500-534, No. 557 (June 2005).
- S. DI ADDARIO and E. PATACCHINI, *Wages and the city. Evidence from Italy*, Labour Economics, v.15, 5, pp. 1040-1061, **TD No. 570 (January 2006).**
- M. PERICOLI and M. TABOGA, Canonical term-structure models with observable factors and the dynamics of bond risk premia, Journal of Money, Credit and Banking, v. 40, 7, pp. 1471-88, TD No. 580 (February 2006).
- E. VIVIANO, *Entry regulations and labour market outcomes. Evidence from the Italian retail trade sector*, Labour Economics, v. 15, 6, pp. 1200-1222, **TD No. 594 (May 2006).**
- S. FEDERICO and G. A. MINERVA, Outward FDI and local employment growth in Italy, Review of World Economics, Journal of Money, Credit and Banking, v. 144, 2, pp. 295-324, TD No. 613 (February 2007).
- F. BUSETTI and A. HARVEY, *Testing for trend*, Econometric Theory, v. 24, 1, pp. 72-87, **TD No. 614** (February 2007).
- V. CESTARI, P. DEL GIOVANE and C. ROSSI-ARNAUD, *Memory for prices and the Euro cash changeover: an analysis for cinema prices in Italy*, In P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, **TD No. 619 (February 2007)**.
- H. BRONWYN, F. LOTTI and J. MAIRESSE, Employment, innovation and productivity: evidence from Italian manufacturing microdata, Industrial and Corporate Change, v. 17, 4, pp. 813-839, TD No. 622 (April 2007).
- J. SOUSA and A. ZAGHINI, *Monetary policy shocks in the Euro Area and global liquidity spillovers*, International Journal of Finance and Economics, v.13, 3, pp. 205-218, **TD No. 629 (June 2007).**

- M. DEL GATTO, GIANMARCO I. P. OTTAVIANO and M. PAGNINI, Openness to trade and industry cost dispersion: Evidence from a panel of Italian firms, Journal of Regional Science, v. 48, 1, pp. 97-129, TD No. 635 (June 2007).
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "inflation perceptions"? A survey-based analysis of Italian consumers, in P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 655 (January 2008).
- B. BORTOLOTTI, and P. PINOTTI, *Delayed privatization*, Public Choice, v. 136, 3-4, pp. 331-351, **TD No.** 663 (April 2008).
- R. BONCI and F. COLUMBA, *Monetary policy effects: New evidence from the Italian flow of funds*, Applied Economics, v. 40, 21, pp. 2803-2818, **TD No. 678 (June 2008).**
- M. CUCCULELLI, and G. MICUCCI, *Family Succession and firm performance: evidence from Italian family firms*, Journal of Corporate Finance, v. 14, 1, pp. 17-31, **TD No. 680 (June 2008).**

2009

- S. MAGRI, *The financing of small innovative firms: The Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, Nonlinearities in the dynamics of the euro area demand for M1, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**

#### FORTHCOMING

- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, **TD No. 458 (December 2002).**
- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD** No. 578 (February 2006).
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, **TD No. 597 (September 2006).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, **TD No. 601 (September 2006).**
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, **TD No. 637 (June 2007).**
- L. FORNI, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: Estimates for the euro area*, Journal of Public Economics, **TD No. 652 (November 2007).**
- R. GOLINELLI and S. MOMIGLIANO, *The cyclical reaction of fiscal policies in the Euro Area. A critical survey of empirical research*, Fiscal Studies, **TD No. 654 (January 2008).**
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Regional Science, **TD No. 688** (September 2008).